Author's response to reviews

Title: Neuroprotective effect of Tagara—an ayurvedic drug against oxidative stress induced by methyl mercury toxicity in rat brain mitochondrial enriched fractions

Authors:

Dhanoop Manikoth Ayyathan (dhanoop@vit.ac.in)
Rajasekaran Chandrasekaran (drcrs70@gmail.com)
Kalaivani Thiagarajan (kalaivanit@vit.ac.in)

Version: 2 Date: 2 July 2014

Author's response to reviews:

To

Editor-in-chief,

BMC Complementary and Alternative medicine.

Sub: Submission of revised manuscript entitled “Neuroprotective Effect of Tagara—an Ayurvedic drug against Oxidative Stress Induced by Methyl Mercury Toxicity in Rat Brain Mitochondrial Enriched Fractions” - reg.

Dear Sir,

I hereby submit a revised manuscript entitled “Neuroprotective Effect of Tagara—an Ayurvedic Drug against Oxidative Stress Induced by Methyl Mercury Toxicity in Rat Brain Mitochondrial Enriched Fractions” to be considered for publication as a research article in BMC Complementary and Alternative medicine.

As MeHg is a health hazard that causes neurological disorders. Till date no proper drug has been found. The main aim of our research is to study the protective effect of Tagara an ayurvedic drug and analyzing its significance. This article reveals the molecular mechanism of action of Tagara against methyl mercury induced oxidative stress. We believe these findings will be of interest to the scholars working in this area.

We declare that this manuscript is original, has not been published before and is not currently being considered for publication elsewhere.

We wish to confirm that there are no known conflicts of interest associated with this publication and there has been no significant financial support for this work that could have influenced its outcome.
"All other Authors have read the manuscript and have agreed to submit it in its current form for consideration for publication in the Journal."

# Line numbers and page numbers are included in the revised manuscript as advised.

We hope you find our manuscript suitable for publication work and look forward to hearing from you.

Kindly send our manuscript to appropriate reviewers and oblige.

Sincerely,

Dr. T. Kalaivani,
Associate Professor (Senior),
Biomedical Sciences Division,
School of Biosciences and Technology,
VIT University,
Vellore-632014,
INDIA.